Author's response to reviews

Title: The changing epidemiology of pediatric aseptic meningitis in Daejeon, Korea from 1987 to 2003

Authors:

Kyung-Yil Lee (leekyungyil@catholic.ac.kr)
David Burgner (dburgner@ichr.uwa.edu.au)
Hyung-Shin Lee (vitalina@hanmail.net)
Ja-Hyun Hong (nelson@mdhouse.com)
Mi-Hee Lee (lpedleemj@catholic.ac.kr)
Jin-Han Kang (kijhan@olmh.cuk.ac.kr)
Byung-Churl Lee (byungcl@catholic.ac.kr)

Version: 9 Date: 10 September 2005

Author's response to reviews: see over
Dear Dr. Newmark

We appreciate again your kind consideration for our manuscript (#MS: 5364325606076875). We are sorry to make a mistake in formatting of figures. We cropped our manuscript’s figures as closely as possible to minimize white space around the image.

Sincerely Yours,

Kyung-Yil Lee, MD
Department of Pediatrics,
The Catholic University of Korea, Daejeon St. Mary's Hospital,
520-2 Daheung-dong, Jung-gu, Daejeon 301-723, Republic of Korea
Tel: +82-42-220-9541, Fax: +82-42-221-2925,
E-mail: leekyungyil@catholic.ac.kr